MCC Operations Status Meeting

Tuesday,  January 29, 2008
This afternoons 4:15pm meeting:
Everyone:

Software Day scheduled for tomorrow, Wednesday January 30, 2008 from 8am-4pm.

The schedule is posted on the LCLS wiki (see below). Postponed Software Day tasks are the MP00 and PX00 reboot.
http://lcls-dev/tiki-index.php?page=SoftwareDaySchedule_30-Jan-2008
Aida: Greg White
1. Server went down.

BPMS:  Operations

For the replaced bpm CAMAC modules in li28 and soon in li29, the bpm’s have a new public bpm calibration from the SCP will need to be done for each.
Profile Monitors: Sheng Peng and Zen Szalata
1.  The TGT pv is not restored on ioc reboot. For example, if the TGT is IN, on reboot it is OUT. 

This morning at the 8:00am meeting:

SCORE: Debbie Rogind

1. Atrimis # 82560. Operators were only able to restore do partial Magnet BCON restores.  Debbie is looking into this problem. 
Magnets: Kristi Luchini

1. Unable to trim QE03.  Kristi is looking the log to determine why the operator was not able to trim last night. However, is QE03 showed trimmed. 
BPMS: Karey Krauter and Till Strauman

1. General comment, that the physicists want to make sure that the 3 new bc2 bpms will have their EPICS db setup with good default values at the start of commissioning.
